Author’s response to reviews

Title: Percutaneous therapy of a mediastinal lymphangioma with fibrin glue: case report with clinical success after 4 years

Authors:

Siyu Zhou (zgydmay@163.com)
Siyuan Dong (dsycmu@163.com)
Jiang Du (dujiangcmu@163.com)

Version: 2 Date: 29 Nov 2017

Author’s response to reviews:

Dear editor and reviewers: Thank you for your letter and for the editors’ and reviewer’s comments concerning our manuscript entitled “Percutaneous therapy of a mediastinal lymphangioma with fibrin glue: case report with clinical success after 4 years”(ID: BSUR-D-17-00329R1). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our research. We have studied comments carefully and have made correction which we hope meet with approval. The main corrections in the paper and the responds to the editors’ and reviewer’s comments are as flowing.Francesco Pia (Reviewer 1): Francesco Pia (Reviewer 1): Include in the text the comment about small cysts like this :" We infer that the multiple small cysts might communicate with the biggest one, because after drainage, nearly all the small cysts were disappeared" Thanks for your suggestions. And we have add this sentence into our manuscript.Mentionin the follow-up description like you explain to us:"This patient came to our hospital to review this year. At present the patient is remain asymptomatic. And she taken a CT at other hospital this year which manifested no recurrence(the same with the CT in this article). Regrettable, because the CT was not taken in our hospital, we can not got a clear image which can be published in the article so we use the CT images she taken 1.5 year after treatment. When she come to our hospital next time I will try to get a clear CT in our hospital." Thanks for your valuable suggestions. And we have add this section into our manuscript. The readers may be confused without this part.